%~MoulinÃ© par MaN_auto v.0.33.8 (9c3f6e85) 2025-05-09 10:05:17
\documentclass[CRMATH,Unicode,XML,biblatex]{cedram}

\TopicFR{ProbabilitÃ©s}
\TopicEN{Probability theory}

\addbibresource{CRMATH_Tien-Dung_20240715.bib}

\usepackage{dsfont}
\usepackage{mathtools}

\newcommand*\ind{\mathds{1}}

\newcommand*{\dd}{\mathrm{d}}

\DeclareMathOperator{\Var}{Var}
\DeclareMathOperator{\Cov}{Cov}
\DeclareMathOperator{\Dom}{Dom}

%%%---------------------------------------------------------------------------

%%% DELIMITERS

%\usepackage{mathtools}
\DeclarePairedDelimiter{\parens}{\lparen}{\rparen}
\DeclarePairedDelimiter{\braces}{\{}{\}}
\DeclarePairedDelimiter{\abs}{\lvert}{\rvert}
\DeclarePairedDelimiter{\norm}{\lVert}{\rVert}
\DeclarePairedDelimiter{\bracks}{[}{]}
\DeclarePairedDelimiter{\angles}{\langle}{\rangle}
\DeclarePairedDelimiter{\aangles}{\langle\!\langle}{\rangle\!\rangle}
\DeclarePairedDelimiter{\pparens}{(\!(}{)\!)}

%%%---------------------------------------------------------------------------

%%% Env. {step}
\newcounter{step}
\newenvironment{step}[1][]{\refstepcounter{step}\begin{proof}[{Step~\thestep\if#1\empty\else: #1\fi}]}{\let\qed\relax\end{proof}}

%%% Env. {case} avec rÃ©fÃ©rences (les pointeurs sont faussÃ©s sinon)
\theoremstyle{remark}
\newcounter{setofcases}
\newcounter{case}
\newenvironment{case}[1][]{\refstepcounter{case}\begin{proof}[{Case~\thecase\if#1\empty\else : \boldmath{#1}\fi}]}{\let\qed\relax\end{proof}}
%\renewcommand*{\thecase}{\Roman{case}}
\newcommand{\newcaseset}{\stepcounter{setofcases}}

%%%---------------------------------------------------------------------------

\graphicspath{{./figures/}}

\newcommand*{\mk}{\mkern -1mu}
\newcommand*{\Mk}{\mkern -2mu}
\newcommand*{\mK}{\mkern 1mu}
\newcommand*{\MK}{\mkern 2mu}

\hypersetup{urlcolor=purple, linkcolor=blue, citecolor=red}

\let\oldtilde\tilde
\renewcommand*{\tilde}[1]{\mathchoice{\widetilde{#1}}{\widetilde{#1}}{\oldtilde{#1}}{\oldtilde{#1}}}
\let\oldforall\forall
\renewcommand*{\forall}{\mathrel{\oldforall}}

%%%---------------------------------------------------------------------------

%%% Changer la largeur des adresses

% Jouer avec \hsize :
\makeatletter
\def\@setafterauthor{%
  \vglue3mm%
%  \hspace*{0pt}%
\begingroup\hsize=11.1cm\advance\hsize\abstractmarginL\raggedright
\noindent
%\hspace*{\abstractmarginL}\begin{minipage}[t]{10cm}
   \leftskip\abstractmarginL
  \normalfont\Small
  \@afterauthor\par
\endgroup
\vskip2pt plus 3pt minus 1pt
}
\makeatother

%%%---------------------------------------------------------------------------

\title{Non-uniform Berry--Esseen bounds via Malliavin--Stein method}

\alttitle{Limites de Berry--Esseen non uniformes par la mÃ©thode de Malliavin--Stein}

\author{\firstname{Nguyen} \lastname{Tien Dung}\IsCorresp}
\address{Department of Mathematics, VNU University of Science, Vietnam National University, Hanoi, 334 Nguyen Trai, Thanh Xuan, Hanoi, 084 Vietnam}
\email{dung@hus.edu.vn}

\author{\firstname{Le} \lastname{Vi}}
\address[1]{Department of Mathematics, VNU University of Science, Vietnam National University, Hanoi, 334 Nguyen Trai, Thanh Xuan, Hanoi, 084 Vietnam}
\email{levi121286@gmail.com}

\author{\firstname{Pham Thi} \lastname{Phuong Thuy}}
\address{The faculty of Basic Sciences, Vietnam Air Defence and Air Force Academy, Son Tay, Hanoi, 084 Vietnam}
\email{thuypham82.com@gmail.com}

\begin{abstract}
In this paper, we establish non-uniform Berry--Esseen bounds by means of the Malliavin--Stein method. Applications to the multiple Wiener--It\^o integrals and the exponential functionals of Brownian motion are given to illustrate the theory.
\end{abstract}

\begin{altabstract}
Dans cet article, nous Ã©tablissons des bornes de Berry--Esseen non uniformes au moyen de la mÃ©thode de Malliavin--Stein. Des applications aux intÃ©grales multiples de Wiener--It\^o et aux fonctions exponentielles du mouvement brownien sont donnÃ©es pour illustrer la thÃ©orie.
\end{altabstract}

\keywords{\kwd{Malliavin--Stein method} \kwd{non-uniform Berry--Esseen bound}}
\altkeywords{\kwd{MÃ©thode de Malliavin--Stein} \kwd{limite de Berry--Esseen non uniforme}}

\subjclass{60F05, 60G15, 60H07}

\thanks{This research was funded by Viet Nam National Foundation for Science and Technology Development (NAFOSTED) under grant number~101.03-2021.29}
\CDRGrant[Viet Nam National Foundation for Science and Technology Development (NAFOSTED)]{101.03-2021.29}

\begin{document}

\maketitle

\section{Introduction}

The Malliavin--Stein method was introduced by Nourdin and Peccati in 2009. Their seminal paper~\cite{NourdinPe2009} combines Stein's method and Malliavin calculus to obtain quantitative central limit theorems for the functionals of a Gaussian process. Since then, many important achievements have been obtained by various authors. For a long list of papers using the Malliavin--Stein method, the reader can consult the constantly updated website~\url{https://sites.google.com/site/malliavinstein/home}.

One of the main results obtained in~\cite{NourdinPe2009} is the following uniform Berry--Esseen bound: let $Z\sim N(0,1)$ and let $F\in \mathbb{D}^{1,2}$ with zero mean, if the law of $F$ is absolutely continuous with respect to the Lebesgue measure, then we have
\begin{equation}\label{whjo}
\sup_{z\in \mathbb{R}}\abs[\big]{P(F\leq z)-P(Z\leq z)}\leq \sqrt{E\abs[\big]{1-\angles{DF,-DL^{-1}F}_{\mathfrak{H}}}^2}.
\end{equation}
The space $\mathbb{D}^{1,2}$, the operators $D$ and $L^{-1}$ will be defined in Section~\ref{uufk} below. We recall that the uniform Berry--Esseen bounds are very useful in statistical applications. However, such bounds only work well for values of $z$ near the center of the distribution. For $z$ sufficiently large, the difference $P(F\leq z)-P(Z\leq z)$ becomes so close to zero and the uniform bounds are too crude to be of any use.

In this paper, our aim is to obtain the non-uniform Berry--Esseen bounds standing for~\eqref{whjo}. More specifically, under the same assumption as in~\cite{NourdinPe2009}, we obtain in Theorem~\ref{kfoi} below the following bound
\begin{equation}\label{h4jo}
\abs[\big]{P(F\leq z)-P(Z\leq z)}\leq \sqrt{E\abs[\big]{1-\angles{DF,-DL^{-1}F}_{\mathfrak{H}}}^2}\parens[\Bigg]{\sqrt{P\parens[\big]{\abs{F}>\abs{z}/2}}+2e^{-z^2/4}} \quad \forall z\in \mathbb{R}.
\end{equation}
In the classical central limit theorem for independent random variables, the rate of convergence with respect to $z$ is of order $O\parens[\big]{1+\abs{z}^3}$. Here, by using Markov's inequality, our bound~\eqref{h4jo} also gives the same rate whenever $E\abs{F}^6$ is finite. Furthermore, we can employ the theory of concentration inequalities to evaluate $P\parens[\big]{\abs{F}>\abs{z}/2}$ and, in many situations, we are able to obtain a much faster rate. For example, when applied to $F=I_q(f)$ the multiple Wiener--It\^o integral of order $q\geq 2$, the bound~\eqref{h4jo} gives us an exponential rate that reads
\[
\abs[\big]{P(F\leq z)-P(Z\leq z)}\leq \sqrt{\frac{q-1}{3q}\parens[\big]{E\abs{F}^4-3}}\parens[\big]{c_qe^{-\frac{z^{2/q}}{2^{2+2/q}}}+2e^{-\frac{z^2}{4}}} \quad \forall z\in \mathbb{R},
\]
where $c_q$ is a positive constant.

The rest of this article is organized as follows. In Section~\ref{uufk}, we briefly recall some of the relevant elements of the Malliavin calculus. Our main results are then formulated and proved in Section~\ref{uloq}. In Section~\ref{9jko}, we apply our results to derive non-uniform Berry--Esseen bounds for the multiple Wiener--It\^o integrals and the exponential functionals of Brownian motion.

\section{Malliavin Calculus}\label{uufk}

Let us recall some elements of Malliavin calculus that we need in order to perform our proofs (for more details see~\cite{nualartm2}). Suppose that $\mathfrak{H}$ is a real separable Hilbert space with scalar product denoted by $\angles{\,\cdot\,,\cdot\,}_\mathfrak{H}$. We denote by $W = \braces[\big]{W(h) : h \in \mathfrak{H}}$ an isonormal Gaussian process defined in a complete probability space $(\Omega,\mathcal{F},P)$, $\mathcal{F}$ is the $\sigma$-field generated by $W$. Let $\mathcal{S}$ be the set of all smooth cylindrical random variables of the form
\begin{equation}\label{ro}
F=f\parens[\big]{W(h_1),\dotsc,W(h_n)},
\end{equation}
where $n\in \mathbb{N}, f\in C_b^\infty(\mathbb{R}^n)$ the set of bounded and infinitely differentiable functions with bounded partial derivatives, $h_1,\dotsc,h_n\in \mathfrak{H}$. If $F$ has the form~\eqref{ro}, we define its Malliavin derivative with respect to $W$ as the element of $L^2(\Omega,\mathfrak{H})$ given by
\[
DF=\sum_{k=1}^n \frac{\partial f}{\partial x_k}\parens[\big]{W(h_1),\dotsc,W(h_n)} h_k.
\]
More generally, we can define the $k$th order derivative $D^kF\in L^2(\Omega, \mathfrak{H}^{\otimes k})$ by iterating the derivative operator $k$ times. For any integer $k\geq 1$ and any $p\geq 1$, we denote by $\mathbb{D}^{k,p}$ the closure of $\mathcal{S}$ with respect to the norm
\[
\norm{F}^p_{k,p}\coloneqq E\abs{F}^p+\sum_{i=1}^kE\norm{D^i F}^p_{\mathfrak{H}^{\otimes i}}.
\]
An important operator in the Malliavin calculus theory is the divergence operator $\delta$, it is the adjoint of the derivative operator $D$ characterized by
\begin{equation}\label{s44}
E\angles{DF,u}_{\mathfrak{H}}=E\bracks[\big]{F\delta(u)}
\end{equation}
for any $F\in \mathcal{S}$ and $u\in L^2(\Omega,\mathfrak{H})$. The domain of $\delta$ is the set of all processes $u\in L^2(\Omega,\mathfrak{H})$ such that
\[
E\abs[\big]{\angles{DF,u}_{\mathfrak{H}}}\leq C(u)\norm{F}_{L^2(\Omega)},
\]
where $C(u)$ is some positive constant depending only on $u$. Let $F\in \mathbb{D}^{1,2}$ and $u\in \Dom \delta$ such that $Fu\in L^2(\Omega,\mathfrak{H})$. Then $Fu\in \Dom \delta$ and we have the following relation
\[
\delta(Fu)=F\delta(u)-\angles{DF,u}_{\mathfrak{H}},
\]
provided the right-hand side is square integrable.

For any integer $q\geq 0$, we denote by $\mathcal{H}_q$ the $q$th Wiener chaos of $W$. We recall that $\mathcal{H}_0=\mathbb{R}$ and, for any $q\geq 1$, $\mathcal{H}_q$ is the closed linear subspace of $L^2(\Omega)$ generated by the family of random variables $I_q(f)$ where $I_q$ indicates a multiple Wiener--It\^o integral of order $q$ and $f\in \mathfrak{H}^{\odot q}$ (the $q$th symmetric tensor power of $\mathfrak{H}$). It is known that any random variable $F$ in $L^2(\Omega)$ can be expanded into an orthogonal sum of its Wiener chaos:
\[
F=E[F]+\sum_{q=1}^\infty I_q(f_q),
\]
where the series converges in $L^2(\Omega)$ and $f_q\in \mathfrak{H}^{\odot q}$. From this chaos expansion one may define the Ornstein--Uhlenbeck operator $L$ by $LF=\sum_{q=1}^\infty-q I_q(f_q)$ when $F\in \mathbb{D}^{2,2}$ and its pseudo-inverse by $L^{-1}\parens[\big]{F-E[F]}=\sum_{q=1}^\infty-\frac{1}{q} I_q(f_q)$. Hereafter, we write $L^{-1}F$ instead of $L^{-1}\parens[\big]{F-E[F]}$. Note that, for any $F\in L^2(\Omega)$, we have $L^{-1}F\in \Dom L$ and $LL^{-1}F=L^{-1}LF=F-E[F]$. Moreover, the operators $D,\delta$ and $L$ satisfy the following relationship: $F\in \Dom L$ if and only if $F\in \mathbb{D}^{2,2}$ and, in this case,
\begin{equation}\label{hj4}
\delta D F=-LF.
\end{equation}

\section{Non-uniform Berry--Esseen bounds}\label{uloq}

Let $\Phi$ denote the cumulative distribution function of standard normal random variable. We recall that, for each $z\in \mathbb{R}$, the Stein equation
\begin{equation*}
f'(x)-xf(x)=\ind_{\{x\leq z\}}-\Phi(z), \quad x\in \mathbb{R},
\end{equation*}
admits a unique solution $f_z(x)$ given by
\begin{equation}\label{uydk}
f_z(x)=
\begin{cases}
\sqrt{2\pi}e^{x^2/2}\Phi(x)\parens[\big]{1-\Phi(z)}	& \text{if $x\leq z$}, \\
\sqrt{2\pi}e^{x^2/2}\parens[\big]{1-\Phi(x)}\Phi(z) 	& \text{if $x>z$}.
\end{cases}
\end{equation}
The key point in our proof is to establish new estimates for $f_z$ and $f'_z$ as in the next lemma.

\begin{lemm}\label{erf}
Let $z>0$ and let $f_z$ be given by~\eqref{uydk}. Then we have
\begin{gather}
0<f_z(x)\leq \frac{\sqrt{2\pi}}{4}, \quad \abs[\big]{f'_z(x)}\leq 1 \quad \forall x\in \mathbb{R}, \label{m9io1}\\
0<f_z(x)\leq \frac{\sqrt{2\pi}}{2}e^{-z^2/4} \quad \forall \abs{x}\leq z/2 \label{m9q}
\end{gather}
and
\begin{equation}\label{m9io}
\abs[\big]{f'_z(x)}\leq 2e^{-z^2/4} \quad \forall \abs{x}\leq z/2.
\end{equation}
\end{lemm}

\begin{proof}
The estimates in~\eqref{m9io1} are well known, see e.g.~\cite[Lemma~2.3]{Chen2011}. Let us prove~\eqref{m9io}. When $x\leq z$, we have $f_z(x)=\sqrt{2\pi}e^{x^2/2}\Phi(x)\parens[\big]{1-\Phi(z)}$ and hence,
\[
f'_z(x)=\parens[\big]{1-\Phi(z)}\parens[\big]{1+\sqrt{2\pi}xe^{x^2/2}\Phi(x)}.
\]
Fixed $z>0$, we deduce
\[
\abs[\big]{f'_z(x)}\leq \parens[\big]{1-\Phi(z)}\parens[\Bigg]{1+\frac{\sqrt{2\pi}}{2}ze^{z^2/8}}, \quad \abs{x}\leq z/2.
\]
We now observe that
\[
1-\Phi(z)=\int_0^\infty \frac{1}{\sqrt{2\pi}}e^{-(u+z)^2/2}\,\dd u\leq e^{-z^2/2}\int_0^\infty \frac{1}{\sqrt{2\pi}}e^{-u^2/2}\,\dd u=\frac{1}{2}e^{-z^2/2}.
\]
Consequently, we obtain
\[
\begin{split}
\abs[\big]{f'_z(x)}	& \leq \frac{1}{2}e^{-z^2/2}\parens[\big]{1+\sqrt{\pi/2}ze^{z^2/8}} \\
			& \leq \frac{1}{2}e^{-z^2/4}\parens[\big]{1+\sqrt{\pi/2}ze^{-z^2/8}}, \quad \abs{x}\leq z/2.
\end{split}
\]
Furthermore, it is easy to verify that $ze^{-z^2/8}\leq 2e^{-1/2}$, $\forall z>0$. So we conclude that
\[
\abs[\big]{f'_z(x)}\leq \frac{1}{2}e^{-z^2/4}\parens[\big]{1+2\sqrt{\pi/2}e^{-1/2}}<2e^{-z^2/4}, \quad \abs{x}\leq z/2.
\]
This completes the proof of~\eqref{m9io}. The verification of~\eqref{m9q} is trivial. Indeed, we have
\[
f_z(x)=\sqrt{2\pi}e^{x^2/2}\Phi(x)\parens[\big]{1-\Phi(z)}\leq \frac{\sqrt{2\pi}}{2}e^{z^2/8}e^{-z^2/2}\leq \frac{\sqrt{2\pi}}{2}e^{-z^2/4}, \quad \abs{x}\leq z/2.
\]
The proof of the lemma is complete.
\end{proof}

We now are in a position to state and prove the main results of the present paper.

\begin{theo}\label{kfoi}
Let $Z\sim N(0,1)$ and let $F\in \mathbb{D}^{1,2}$. If, in addition, the law of $F$ is absolutely continuous with respect to the Lebesgue measure, then we have
\begin{multline}\label{hjo}
\abs[\big]{P(F\leq z)-P(Z\leq z)} \\
\leq \parens[\Bigg]{\abs[\big]{E[F]}+\sqrt{E\abs[\big]{1-\angles{DF,-DL^{-1}F}_{\mathfrak{H}}}^2}}\parens[\Bigg]{\sqrt{P\parens[\big]{\abs{F}>\abs{z}/2}}+2e^{-z^2/4}} \quad \forall z\in \mathbb{R}.
\end{multline}
\end{theo}

\begin{proof}
We first recall the arguments in the proof of~\cite[Theorem~3.1]{NourdinPe2009}. Note that $F=E[F]-\delta DL^{-1}F$. Then, for any $z\in \mathbb{R}$, by using the Stein's equation and the relationship~\eqref{s44}, we have
\[
\begin{split}
P(F\leq z)-P(Z\leq z)
& = E\bracks[\big]{f'_z(F)}-E\bracks[\big]{f_z(F)F}\\
& = E\bracks[\big]{f'_z(F)}-E\bracks[\big]{f_z(F)\parens[\big]{F-E[F]}}-E\bracks[\big]{f_z(F)}E[F]\\
& = E\bracks[\big]{f'_z(F)}+E\bracks[\big]{f_z(F)\delta DL^{-1}F}-E\bracks[\big]{f_z(F)}E[F]\\
& = E\bracks[\big]{f'_z(F)\parens[\big]{1-\angles{DF,-DL^{-1}F}_{\mathfrak{H}}}}-E\bracks[\big]{f_z(F)}E[F].
\end{split}
\]
\begin{case}[$z\geq 0$]\label{case_1}
We rewrite the above expression as follows
\[
\begin{split}
& P(F\leq z)-P(Z\leq z)\\
& \hspace{2cm}		= E\bracks[\bigg]{f'_z(F)\parens[\big]{1-\angles{DF,-DL^{-1}F}_{\mathfrak{H}}}\ind_{\{\abs{F}\leq z/2\}}} + E\bracks[\bigg]{f'_z(F)\parens[\big]{1-\angles{DF,-DL^{-1}F}_{\mathfrak{H}}}\ind_{\{\abs{F}> z/2\}}}\\
& \hspace{2cm}\quad	- E\bracks[\big]{f_z(F)\ind_{\{\abs{F}\leq z/2\}}}E[F]-E\bracks[\big]{f_z(F)\ind_{\{\abs{F}> z/2\}}}E[F].
\end{split}
\]
Then, by Lemma~\ref{erf}, we get
\[
\begin{split}
& \abs[\big]{P(F\leq z)-P(Z\leq z)}\\
& \hspace{1.8cm}		\leq E\abs[\bigg]{f'_z(F)\parens[\big]{1-\angles{DF,-DL^{-1}F}_{\mathfrak{H}}}\ind_{\{\abs{F}\leq z/2\}}}+E\abs[\bigg]{f'_z(F)\parens[\big]{1-\angles{DF,-DL^{-1}F}_{\mathfrak{H}}}\ind_{\{\abs{F}> z/2\}}}\\
& \hspace{1.8cm}\quad	+ \parens[\bigg]{E\bracks[\big]{f_z(F)\ind_{\{\abs{F}\leq z/2\}}}+E\bracks[\big]{f_z(F)\ind_{\{\abs{F}> z/2\}}}}\abs[\big]{E[F]}\\
& \hspace{1.8cm}		\leq 2e^{-z^2/4}E\abs[\big]{1-\angles{DF,-DL^{-1}F}_{\mathfrak{H}}}+E\bracks[\bigg]{\abs[\big]{1-\angles{DF,-DL^{-1}F}_{\mathfrak{H}}}\ind_{\{\abs{F}> z/2\}}}\\
& \hspace{1.8cm}\quad	+\parens[\Bigg]{\frac{\sqrt{2\pi}}{2}e^{-z^2/4}+\frac{\sqrt{2\pi}}{4}P\parens[\big]{\abs{F}>z/2}}\abs[\big]{E[F]}.
\end{split}
\]
So, by applying the Cauchy--Schwarz inequality, we obtain
\[
\abs[\big]{P(F\leq z)-P(Z\leq z)}\leq \parens[\Bigg]{\abs[\big]{E[F]}+\sqrt{E\abs[\big]{1-\angles{DF,-DL^{-1}F}_{\mathfrak{H}}}^2}}\parens[\Bigg]{\sqrt{P\parens[\big]{\abs{F}>z/2}}+2e^{-z^2/4}}.
\]
This finishes the proof of~\eqref{hjo} for $z\geq 0$.
\end{case}
\begin{case}[$z< 0$]
This case follows directly from Case~\ref{case_1}. Indeed, we have
\[
\begin{split}
\abs[\big]{P(F\leq z)-P(Z\leq z)}
& = \abs[\big]{P(-F\geq -z)-P(-Z\geq -z)}\\
& = \abs[\big]{P(-F\geq -z)-P(Z\geq -z)}\\
& = \abs[\big]{P(-F\leq -z)-P(Z\leq -z)}\\
& \leq \parens[\Bigg]{\abs[\big]{E[-F]}+\sqrt{E\abs[\big]{1-\angles[\big]{D(-F),-DL^{-1}(-F)}_{\mathfrak{H}}}^2}}\parens[\Bigg]{\sqrt{P\parens[\big]{\abs{-F}>-z/2}}+2e^{-z^2/4}}\\
& = \parens[\Bigg]{\abs[\big]{E[F]}+\sqrt{E\abs[\big]{1-\angles{DF,-DL^{-1}F}_{\mathfrak{H}}}^2}}\parens[\Bigg]{\sqrt{P\parens[\big]{\abs{F}>-z/2}}+2e^{-z^2/4}}.
\end{split}
\]
The proof of the theorem is complete.\qedhere
\end{case}
\let\qed\relax
\end{proof}
\begin{rema}
When $F$ can be represented as a Skorokhod integral: $F=E[F]+\delta(u)$ for some stochastic process $u$, we use the relationship~\eqref{s44} to get
\[
\begin{split}
P(F\leq z)-P(Z\leq z)&=E\bracks[\big]{f'_z(F)}-E\bracks[\big]{f_z(F)\parens[\big]{F-E[F]}}-E\bracks[\big]{f_z(F)}E[F]\\
&=E\bracks[\big]{f'_z(F)}-E\bracks[\big]{f_z(F)\delta(u)}-E\bracks[\big]{f_z(F)}E[F]\\
&=E\bracks[\big]{f'_z(F)\parens[\big]{1-\angles{DF,u}_{\mathfrak{H}}}}-E\bracks[\big]{f_z(F)}E[F].
\end{split}
\]
Hence, with the exact proof of Theorem~\ref{kfoi}, we also have the following non-uniform Berry--Esseen bound
\begin{equation}\label{h5jo}
\abs[\big]{P(F\leq z)-P(Z\leq z)}\leq \parens[\Bigg]{\abs[\big]{E[F]}+\sqrt{E\abs[\big]{1-\angles{DF,u}_{\mathfrak{H}}}^2}}\parens[\Bigg]{\sqrt{P\parens[\big]{\abs{F}>\abs{z}/2}}+2e^{-z^2/4}} \quad \forall z\in \mathbb{R}.
\end{equation}
\end{rema}

\section{Some applications}\label{9jko}

In this section, we provide some applications to illustrate the results obtained in the previous section.

\subsection{Multiple Wiener--It\^o integrals}

We recall that among the applications of the bound~\eqref{whjo} is the fourth moment theorem for the multiple Wiener--It\^o integrals. Here we have the following non-uniform Berry--Esseen bound.
\begin{theo}
Let $q\geq 2$ be an integer, and let $F= I_q(f)$ have variance one. Then, it holds that
\begin{equation}\label{hjo6}
\abs[\big]{P(F\leq z)-P(Z\leq z)}\leq \sqrt{\frac{q-1}{3q}\parens[\big]{E\abs{F}^4-3}}\parens[\big]{c_qe^{-\frac{z^{2/q}}{2^{2+2/q}}}+2e^{-\frac{z^2}{4}}} \quad \forall z\in \mathbb{R},
\end{equation}
where $c_q$ is a positive constant depending only on $q$.
\end{theo}
\begin{proof}
When $F= I_q(f)$, it is known from~\cite{NourdinPe2009} that
\[
\sqrt{E\abs[\big]{1-\angles{DF,-DL^{-1}F}_{\mathfrak{H}}}^2}\leq \sqrt{\frac{q-1}{3q}\parens[\big]{E\abs{F}^4-3}}.
\]
On the other hand, for the multiple Wiener--It\^o integrals with variance one, \cite[Theorem~4.1]{Major2005} provides us the concentration bound
\[
P\parens[\big]{\abs[\big]{I_q(f)}>x}\leq c_q^2\exp\parens[\Bigg]{-\frac{x^{2/q}}{2}}, \quad x\geq 0,
\]
where $c_q$ is a positive constant depending only on $q$.

So the bound~\eqref{hjo6} follows directly from~\eqref{hjo}.
\end{proof}

To our best knowledge, this is the first time that a non-uniform Berry--Esseen bound for the multiple Wiener--It\^o integrals has appeared in the literature. In addition, this non-uniform bound significantly improves the uniform bound obtained in~\cite{NourdinPe2009}:
\[
\sup_{z\in \mathbb{R}}\abs[\big]{P(F\leq z)-P(Z\leq z)}\leq \sqrt{\frac{q-1}{3q}\parens[\big]{E\abs{F}^4-3}}.
\]

\subsection{Exponential functionals}

Let $B=(B_t)_{t\in [0, T]}$ be a standard Brownian motion defined on a complete probability space $(\Omega,\mathcal{F},\mathbb{F},P)$, where $\mathbb{F}= (\mathcal{F}_t)_{t \in [0,T]}$ is a natural filtration generated by $B$. We consider the exponential functional of the form
\begin{equation}\label{yu1}
F_t=\int_0^t e^{as+B_s}\,\dd s, \quad t\in [0, T],
\end{equation}
where $a$ is a real number. It is known that this functional plays an important role in several domains. A lot of fruitful properties of $F_t$ can be found in the literature, see e.g.~\cite{Matsumoto2005a,Matsumoto2005b,Yor2001}. Let us define
\begin{equation}\label{y6u1}
\tilde{F}_t\coloneqq\frac{F_t-m_t}{\sigma_t}, \quad t\in [0, T],
\end{equation}
where $m_t=E[F_t]$ and $\sigma_t^2=\Var(F_t)$. It was proved in~\cite{Dufresne2004} that $\lim_{t\to 0^+}\frac{m_t}{t}=1$, $\lim_{t\to 0^+}\frac{3\sigma_t^2}{t^3}=1$ and
\[
\tilde{F}_t \ \text{converges in distribution to $Z\sim N(0,1)$ as $t\to 0$}.
\]
In the next theorem, we give an explicit estimate for the rate of convergence.

\begin{theo}\label{vnms}
Let $Z\sim N(0,1)$ and let $(\tilde{F}_t)_{t\geq 0}$ be given by~\eqref{y6u1}. Then, for each $t\in (0,T]$, we have
\[
\abs[\big]{P(\tilde{F}_t\leq z)-P(Z\leq z)}\leq \frac{2e^{2at+4t}t^3\sqrt{t}}{\sigma_t^2} \parens*{\exp\parens*{-\frac{\ln^2\parens[\big]{1+\frac{\abs{z}\sigma_t}{2m_t}}}{4t}}+e^{-z^2/16}+2e^{-z^2/4}} \quad \forall z\in \mathbb{R}.
\]
\end{theo}

In order to see the rate of convergence, we note that
\[
\lim_{t\to 0^+}\frac{2e^{2at+4t}t^3}{\sigma_t^2}=6 \qquad \text{and} \qquad \lim_{t\to 0^+}\frac{\ln^2\parens[\big]{1+\frac{\abs{z}\sigma_t}{2m_t}}}{4t}=\frac{z^2}{48}.
\]
Hereafter, the Malliavin derivative operator $D$ is with respect to $B$. In proofs, we use the following covariance formula: for any $F,G\in \mathbb{D}^{1,2}$, we have
\begin{equation}\label{uijr}
\Cov(F,G)=E\bracks[\Bigg]{\int_0^T D_sFE\bracks[\big]{D_sG \big\vert \mathcal{F}_s}\,\dd s}.
\end{equation}
Before proving Theorem~\ref{vnms}, we first provide the concentration inequalities for $\tilde{F}_t$.

\begin{prop}\label{swde}
Let $(\tilde{F}_t)_{t\geq 0}$ be given by~\eqref{y6u1}. Then, for each $t\in (0,T]$, we have
\begin{equation}\label{dk1}
P(\tilde{F}_t\geq x)\leq e^{-\frac{\ln^2(1+x\sigma_t/m_t)}{2t}}, \quad x\geq 0
\end{equation}
and
\begin{equation}\label{dk2}
P(\tilde{F}_t\leq -x)\leq e^{-x^2/2}, \quad x\geq 0.
\end{equation}
\end{prop}

\begin{proof}
Fixed $t\in (0,T]$, we consider the random variable
\[
X_t=\ln\parens[\Bigg]{\int_0^t e^{as+B_s}\,\dd s}.
\]
It is easy to verify that $X_t$ is a Malliavin differentiable random variable and its derivative is given~by
\[
D_\theta X_t=\frac{\int_\theta^t e^{as+B_s}\,\dd s}{\int_0^t e^{as+B_s}\,\dd s}, \quad 0\leq\theta\leq t.
\]
We have $0< D_\theta X_t\leq 1$ and hence,
\[
0<\int_0^t D_\theta X_tE\bracks[\big]{D_\theta X_t \big\vert \mathcal{F}_\theta}\,\dd \theta\leq t \quad \text{a.s.}
\]
This, together with~\cite[Theorem~2.4]{DPT2011}, implies that
\[
P\parens[\big]{X_t-E[X_t]\geq y}\leq e^{-\frac{y^2}{2t}} \quad \forall y\geq 0.
\]
We now observe that $E[X_t]\leq \ln m_t$ by Lyapunov's inequality. Then, for $x\geq 0$, we deduce
\[
\begin{split}
P(\tilde{F}_t\geq x)&=P\parens[\big]{X_t\geq \ln(x\sigma_t+m_t)}\\
&=P\parens[\big]{X_t-E[X_t]\geq \ln(x\sigma_t+m_t)-E[X_t]}\\
&\leq P\parens[\big]{X_t-E[X_t]\geq \ln(x\sigma_t+m_t)-\ln m_t}\\
&\leq e^{-\frac{\ln^2(1+x\sigma_t/m_t)}{2t}}.
\end{split}
\]
This completes the proof of~\eqref{dk1}. In order to prove the relation~\eqref{dk2}, we consider the function
\[
g(\lambda)=E\bracks[\big]{e^{-\lambda \tilde{F}_t}}, \quad \lambda\geq 0.
\]
By using the covariance formula~\eqref{uijr}, we have
\[
g'(\lambda)=-E\bracks[\big]{e^{-\lambda \tilde{F}_t}\tilde{F}_t}=\lambda E\bracks[\big]{e^{-\lambda \tilde{F}_t}\Gamma_t},
\]
where $\Gamma_t$ is defined by
\begin{equation}\label{hhy}
\Gamma_t\coloneqq\int_0^t D_\theta \tilde{F}_tE\bracks[\big]{D_\theta \tilde{F}_t \big\vert \mathcal{F}_\theta}\,\dd \theta.
\end{equation}
Note that $E[\Gamma_t]=\Cov(\tilde{F}_t,\tilde{F}_t)=1$. Then, once again, we use~\eqref{uijr} to get
\[
\begin{split}
g'(\lambda)
& = \lambda E\bracks[\big]{e^{-\lambda \tilde{F}_t}} + \lambda\Cov\parens[\big]{e^{-\lambda \tilde{F}_t},\Gamma_t}\\
& = \lambda g(\lambda)-\lambda^2E\bracks[\Bigg]{e^{-\lambda \tilde{F}_t}\int_0^t D_r\tilde{F}_tE\bracks[\big]{D_r\Gamma_t \big\vert \mathcal{F}_r}\,\dd r}.
\end{split}
\]
We observe that $\int_0^t D_r\tilde{F}_tE\bracks[\big]{D_r\Gamma_t \big\vert \mathcal{F}_r}\,\dd r\geq 0$ a.s. Indeed, we have for $0\leq r$, $\theta\leq t$,
\[
D_\theta \tilde{F}_t=\frac{1}{\sigma_t}\int_\theta^t e^{as+B_s}\,\dd s\geq 0, \qquad D_rD_\theta \tilde{F}_t=\frac{1}{\sigma_t}\int_\theta^t e^{as+B_s}\ind_{[r,t]}(s)\,\dd s\geq 0
\]
and
\[
D_r\Gamma_t=\int_0^t D_rD_\theta \tilde{F}_tE\bracks[\big]{D_\theta \tilde{F}_t \big\vert \mathcal{F}_\theta}\,\dd \theta+\int_r^t D_\theta \tilde{F}_tE\bracks[\big]{D_rD_\theta \tilde{F}_t \big\vert \mathcal{F}_\theta}\,\dd \theta\geq 0.
\]
So it holds that
\[
g'(\lambda)\leq \lambda g(\lambda), \quad \lambda\geq 0.
\]
This gives us
\[
g(\lambda)\leq e^{\frac{\lambda^2}{2}}, \quad \lambda\geq 0.
\]
Fixed $x\geq 0$, by Markovâ€™s inequality, we have
\[
P(\tilde{F}_t\leq -x)\leq e^{-\lambda x}E\bracks[\big]{e^{-\lambda \tilde{F}_t}}=e^{-\lambda x}g(\lambda)\leq e^{-\lambda x+\frac{\lambda^2}{2}}, \quad \lambda\geq 0.
\]
Choosing $\lambda=x$ yields $P(\tilde{F}_t\leq -x)\leq e^{-x^2/2}$. So the relation~\eqref{dk2} is verified.

The proof of the proposition is complete.
\end{proof}

\begin{proof}[Proof of Theorem~\ref{vnms}]
We will carry out the proof in two steps.
\begin{step}[Moment estimates]
By using the H\"older inequality we have
\[
\begin{split}
E\abs[\big]{D_\theta \tilde{F}_t}^4
& = \frac{1}{\sigma_t^4}E\abs*{\int_\theta^t e^{as+B_s}\,\dd s}^4			\\
& \leq \frac{t^3}{\sigma_t^4}\int_\theta^t E\bracks[\big]{e^{4as+4B_s}}\,\dd s			\\
& = \frac{t^3}{\sigma_t^4}\int_\theta^t e^{4as+8s}\,\dd s					\\
& = \frac{t^4}{\sigma_t^4} e^{4at+8t} \quad \forall 0\leq \theta\leq t
\end{split}
\]
and
\[
E\abs[\big]{D_rD_\theta \tilde{F}_t}^4=\frac{1}{\sigma_t^4}E\abs*{\int_\theta^t e^{as+B_s}\ind_{[r,t]}(s)\,\dd s}^4\leq \frac{t^4}{\sigma_t^4} e^{4at+8t} \quad \forall 0\leq r, \ \theta\leq t.
\]
Let $\Gamma_t$ be given by~\eqref{hhy}. We use the Cauchy--Schwarz and Lyapunov inequalities to deduce
\begin{equation}\label{9ol}
\begin{split}
E\abs[\big]{D_r\Gamma_t}^2
& = E\abs*{\int_0^t D_rD_\theta \tilde{F}_tE\bracks[\big]{D_\theta \tilde{F}_t \big\vert \mathcal{F}_\theta}\,\dd \theta+\int_r^t D_\theta \tilde{F}_tE\bracks[\big]{D_rD_\theta \tilde{F}_t \big\vert \mathcal{F}_\theta}\,\dd \theta}^2 \\
& \leq 2t\int_0^t E\abs[\bigg]{D_rD_\theta \tilde{F}_tE\bracks[\big]{D_\theta \tilde{F}_t \big\vert \mathcal{F}_\theta}}^2\dd \theta+2t\int_r^t E\abs[\bigg]{D_\theta \tilde{F}_tE\bracks[\big]{D_rD_\theta \tilde{F}_t \big\vert \mathcal{F}_\theta}}^2\dd \theta \\
& \leq 2t\int_0^t \sqrt{E\abs[\big]{D_rD_\theta \tilde{F}_t}^4E\abs[\big]{D_\theta \tilde{F}_t}^4}\,\dd \theta+2t\int_r^t \sqrt{E\abs[\big]{D_\theta \tilde{F}_t}^4E\abs[\big]{D_rD_\theta \tilde{F}_t}^4}\,\dd \theta \\
& \leq \frac{4t^6}{\sigma_t^4} e^{4at+8t} \quad \forall 0\leq r\leq t.
\end{split}
\end{equation}
\end{step}
\begin{step}[Conclusion]
By the Clark--Ocone formula, we have
\[
\tilde{F}_t=E[\tilde{F}_t]
+\int_0^tE\bracks[\big]{\tilde{F}_t \big\vert \mathcal{F}_\theta}\,\dd B_\theta.
\]
Hence, by using the bound~\eqref{h5jo} with $u_\theta=E\bracks[\big]{\tilde{F}_t \big\vert \mathcal{F}_\theta}$, we obtain
\begin{equation}
\abs[\big]{P(\tilde{F}_t\leq z)-P(Z\leq z)}\leq \sqrt{E\abs[\big]{1-\Gamma_t}^2}\parens[\Bigg]{\sqrt{P\parens[\big]{\abs{\tilde{F}_t}>\abs{z}/2}}+2e^{-z^2/4}} \quad \forall z\in \mathbb{R}.
\end{equation}
Thanks to Proposition~\ref{swde} we have
\[
\begin{split}
P\parens[\big]{\abs{\tilde{F}_t}>\abs{z}/2}
& = P\parens[\big]{\tilde{F}_t>\abs{z}/2} + P\parens[\big]{\tilde{F}_t<-\abs{z}/2}\\
& \leq \exp\parens*{-\frac{\ln^2\parens[\big]{1+\frac{\abs{z}\sigma_t}{2m_t}}}{2t}}+e^{-z^2/8} \quad \forall z\in \mathbb{R}.
\end{split}
\]
On the other hand, recalling~\eqref{9ol}, we have
\[
E\abs[\big]{1-\Gamma_t}^2=\Cov(\Gamma_t,\Gamma_t)\leq \int_0^t E\abs[\big]{D_r\Gamma_t}^2\,\dd r\leq \frac{4t^7}{\sigma_t^4} e^{4at+8t}.
\]
Combining the above computations yields
\[
\abs[\big]{P(\tilde{F}_t\leq z)-P(Z\leq z)}\leq \frac{2e^{2at+4t}t^3\sqrt{t}}{\sigma_t^2} \parens*{\exp\parens*{-\frac{\ln^2\parens[\big]{1+\frac{\abs{z}\sigma_t}{2m_t}}}{4t}}+e^{-z^2/16}+2e^{-z^2/4}} \quad \forall z\in \mathbb{R}.
\]
This finishes the proof of Theorem~\ref{vnms}.\qedhere
\end{step}
\let\qed\relax
\end{proof}

\section*{Declaration of interests}

The authors do not work for, advise, own shares in, or receive funds from any organization that could benefit from this article, and have declared no affiliations other than their research organizations.

\printbibliography

\end{document}